Author’s response to reviews

Title: A novel compound heterozygous mutation of the SMARCAL1 gene leading to mild Schimke immune-osseous dysplasia: A case report

Authors:

Shuaimei Liu (719305071@qq.com)
Mingchao Zhang (zmchj99@163.com)
Mengxia Ni (nimengxia@126.com)
Peiran Zhu (980346785@qq.com)
Xinyi Xia (xiayjnju@163.com)

Version: 3 Date: 14 Aug 2017

Author’s response to reviews:

Dear editor:

I have received your comments. Thank you for your hard work. The manuscript has been carefully revised according to the comments.

Editor Comments:

1. Ethics and consent to participate AND Consent to publish:

In these sections you mention that consent was obtained for the patients enrolled in the study. However, you only report a singular case.

Please clarify this statement and amend accordingly.
Since the patient in under aged, the consent should have come from the guardians / parents.

Answer 1: There includes the above mentioned amendments and other changes.

The modification of Consent for publication: “Informed written consent was obtained from all the participants for publication of their clinical details and/or clinical images.” has been changed to “Informed written consent was obtained from the patient’s parents for publication of this case report and any accompanying images.”

The modification of Ethics approval and consent to participate: “An informed consent for publication was also obtained from the individuals included in the submission” has been changed to “An informed consent for publication was also obtained from the patient’s parents included in the submission”

Best regards

Xinyi Xia

Department of Centre Laboratory
Institute of Clinical Laboratory Medicine
Jingling Hospital, Nanjing University School of Medicine
NO.305 Zhongshan Road, Xuanwu District
Nanjing, 210002, P.R.China
Tel: +86 025 80863084